Author's response to reviews

Title: A Multi-center, Randomized, Clinical Study to Compare the Effect and Safety of Autologous Cultured Osteoblast Injection to Treat Fractures

Authors:

Seok-Jung Kim (peter@catholic.ac.kr)
Yong-Woon Shin (woonysos@daum.net)
Kyu-Hyun Yang (kyang@vume.yonsei.ac.kr)
Sang-Bum Kim (sangbume@hitel.net)
Moon-Jib Yoo (mijoo@dku.edu)
Suk-Ku Han (hnsukku@catholic.ac.kr)
Soo-Ah Im (saim@catholic.ac.kr)
Yoo-Dong Won (yoodong1@catholic.ac.kr)
Yerl-Bo Sung (ybs58@sanggyepaik.ac.kr)
Taek-Soo Jeon (sirjeon@kyuh.co.kr)
Cheong-Ho Chang (chk@swcell.com)
Jae-Deog Jang (idjang@swcell.com)
Hyun-Jo Kim (angel@swcell.com)
Su-Young Lee (sushine@swcell.com)
Sae-Bom Lee (saebom@swcell.com)

Version: 15 Date: 2 January 2009

Author's response to reviews:

However, before acceptance, could you refer to the new figures in the text, or potentially add them as supplementary data. There are also some formatting issues that you will need to address.

--> I refered to new figures in the text.

Abstract - Please could you structure your abstract according to the guidelines provided at this page: http://www.biomedcentral.com/info/ifora/abstracts

--> I changed the structure of my abstract accorging to the guideline.

The trial registration number should also be included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Please note that there should be no space between the letters and numbers of the trial registration number.

--> I added it in the last part of abstract as you recommanded.

Trial registration: Current Controlled Trials ISRCTN10637905

Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe
the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

--> I mentioned the source of finding of this study in acknowledgement section.

Please also note that manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

--> We changed it according to the guide line.